Author's response to reviews

Title: Beyond Bricks and Mortar: A Rural Network Approach to Preclinical Medical Education

Authors:

Douglas L Myhre (dmyhre@ucalgary.ca)
Paul Adamiak (pjadamia@ucalgary.ca)
Nathan Turley (nturley@ucalgary.ca)
Ron Spice (rnspice@ucalgary.ca)
Wayne Woloschuk (woloschu@ucalgary.ca)

Version: 3
Date: 15 July 2014

Author's response to reviews:

In accordance with the editors request we have completed the following revisions:

1. The funding information has been moved from the "Disclosure of funding" section on the title page to immediately follow the "Acknowledgements" section at the end of the article.

2. The "Disclaimer" and "Other Disclosures" sections at the end of the article have been removed.

3. All ethics approval information has been moved into the "Methods" section.